Author’s response to reviews

Title: Inherited thrombotic thrombocytopenic purpura mimicking Immune Thrombocytopenic Purpura: a case report during pregnancy

Authors:

VALTER ROMÃO DE SOUZA JUNIOR (jr_walter@hotmail.co.uk)
Ana Beatriz Cavalcante de Oliveira (anabeatrizoliveira22@hotmail.com)
Ana Maria Vanderlei (manuhazin@yahoo.com)
Amanda Queiroz da Mota Silveira Aroucha (amanda.motasilveira@gmail.com)
Bruna Duarte (brunapduarte@gmail.com)
Aureli Nunes Machado (aurelinnm@gmail.com)
Lívia Netto Chaer (livianettochaer@gmail.com)
Cláudia Wanderley de Barros Correia (claudia.wanderley@uol.com.br)
Maria da Conceição de Barros Correia (cbarroscorreia@yahoo.com.br)
Manuela Freire Hazin Costa (manuhazin@yahoo.com.br)

Version: 3 Date: 08 Dec 2017

Author’s response to reviews:

1. At the end of the Introduction we added a paragraph describing why this case is presented.

2. we Added results of Complete physical and neurological examination on admission. we gave the name dosage of the steroid treatment.

3. we gave a followup results for a least 6 months.

4. we Discussed what is unique to this case compared with the literature.